Author's response to reviews

**Title:** Implication of Xenobiotic Metabolizing Enzyme gene (CYP2E1, CYP2C19, CYP2D6, mEH and NAT2) Polymorphisms in Breast Carcinoma.

**Authors:**

Achraf Khedhaier (a_khedhaier@yahoo.fr)
Elham Hassen (elham_tn@yahoo.fr)
Noureddine Bouaouina (noureddine.bouaouina@rns.tn)
Sallouha Gabbouj (sallouha_tn@yahoo.fr)
Slim Ben ahmed (slim.benahmed@rns.tn)
Lotfi Chouchane (lotfi.chouchane@planet.tn)

**Version:** 6  **Date:** 17 April 2008

**Author's response to reviews:** see over
April 17th, 2008

Dr Maria Kowalczuk

Senior Assistant Editor, BMC-series journals

Manuscript ID : 1231888224155443
Manuscript title: Implication of Xenobiotic Metabolizing Enzyme gene (CYP2E1, CYP2C19, CYP2D6, mEH and NAT2) Polymorphisms in Breast Carcinoma.

Dear Dr Maria Kowalczuk,

We thank you so much for your mails. Please find below a table which listed all the changes (red underlined) done in this revised version. The changes were made for the References section and for table 4.

<table>
<thead>
<tr>
<th>Page</th>
<th>Line</th>
<th>Section</th>
<th>Latest version</th>
<th>Revision in this version</th>
</tr>
</thead>
<tbody>
<tr>
<td>21</td>
<td>2</td>
<td>References</td>
<td>…JH, Schaapveld M, de Vries EG . Genes other than BRCA1…</td>
<td>…JH, Schaapveld M, de Vries EG . Genes other than BRCA1…</td>
</tr>
<tr>
<td>23</td>
<td>4</td>
<td>References</td>
<td>Urinary 8-hydroxy-2'-deoxyguanosine</td>
<td>Urinary 8-hydroxy-2’-deoxyguanosine</td>
</tr>
<tr>
<td>23</td>
<td>17</td>
<td>References</td>
<td>Microsomal</td>
<td>Microsomal</td>
</tr>
<tr>
<td>28</td>
<td>4</td>
<td>Table 4 (last colomn)</td>
<td>2.82(0.58-18 57)</td>
<td>2.82(0.58-18.57)</td>
</tr>
<tr>
<td>28</td>
<td>10</td>
<td>Table 4 (last colomn)</td>
<td>3.37(0.45-29 24)</td>
<td>3.37(0.45-29.24)</td>
</tr>
<tr>
<td>28</td>
<td>16</td>
<td>Table 4 (last colomn)</td>
<td>1.04(065-1 67)</td>
<td>1.04(065-1.67)</td>
</tr>
</tbody>
</table>

Please, if you have further requirements, do not hesitate to contact us. Thank you.

Yours sincerely,
Dr Elham HASSEN